VC- /Week 50 starting   18Dec06
SM18 Operation Meeting (Tuesdays) 19 dec  06

Bench status  on monday 18-Dec-2006 at   12/18/2006 2:56:26 PM 
	2
	A1
	MBAR2411
	PT 6.1 Training 1
	1.90

	7
	A2
	MBAR2416
	PT 14 Magnet MAPS
	295.18

	5
	B1
	SSS519
	ICS 5 Traveller Last
	296.88

	1
	B2
	SSS516
	PT1 IAP@1.9K, Part III
	1.90

	8
	C1
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	295.09

	3
	C2
	SSS370
	Thermal Cycle
	2.88 HV

	4
	D1
	SSS669
	PT 6.9 Training 9
	14.47

	6
	D2
	SSS695
	PT 13.2 Resist. Meas.
	295.09

	9
	E1
	MBAR2402
	Prep 5 Cool Down
	239.86

	10
	E2
	MBBR2394
	Prep 3 CDW1 HV
	294.73

	11
	F1
	MBBR1345
	PT 14 Magnet MAPS
	294.55

	12
	F2
	MBAL1406
	PT 14 Magnet MAPS
	295.89


CONFIG :  D 6 - AS 2 -  IRSSS 4 ;   
Supervisor:  Vittorio ;        Eq support 16-3602 : 

Last Quench on Thursday 21st Dec at 1700 hrs ..  Shutdown Friday evening 22nd ( no night shift)
10 Magnet Tests completed ( 7 Magnet Stripped ) ; 

5 Dipoles tests


2 Arc-SSS (  365, 368 )

1 500-series (  525 (245 hrs) )

2  600-series [ 674( 187 hrs)   694 (311 hrs)]
 Situation from Cryogenics:  J-P Lamboy .  
General Points:

· Situation SSSS  :D1 & D2 , B1 B2 issues J Halik : start situation  8 Jan ?
· PUT 509 on B ..   any other 5-series ?
· Emptying D cluster completely before Friday 22nd ?
· Eq Support Issues last Sat night & HWREC ..  (reboot like W-XP ? , no solution in 3 yrs?)
· A.O.B
TERMINATION TICK OFF COUNTER 

 ( SPECIAL & NORMAL SSS )

	MAGNET
	ARRIVAL
	DEPARTURE
	 Tick if Done 

	525
	05/12/2006 
	15 dec 06
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	509
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	523
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	516
	15 dec 06
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	519
	 
08/12/2006 


	Waiting to go
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	529
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	530
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	531
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	669
	14 dec 06
	Waiting to go
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	695
	11/12/2006 
	Waiting to go 
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	370
	12 dec 06
17Dec :ROBTC
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	099
	18/12/2006
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Updated :  December 20, 2006
TERMINATION TICK OFF COUNTER 

 ( DIPOLES )
	MAGNET
	ARRIVAL
	DEPARTURE
	TICK IF DONE

	1345
	10-12-2006
	Waiting to Depart
	[image: image18.wmf]DONE



	2263
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	2267
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	2268
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	2326
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	2328
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	2417
	09-12-2006
	16-12-2006
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	2418
	
	
	[image: image25.wmf]DONE



	2419
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	2420
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	1406
	12-12-2006
	Waiting to Depart
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	2363
	30-11-2006
	13-12-2006
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	2394
	16-12-2006
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	2399
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	2402
	16-12-2006
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	2407
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	2408
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	2411
	13-12-2006
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	2414
	
	
	[image: image36.wmf]DONE



	2415
	07-12-2006
	13-12-2006
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	2416
	05-12-2006
	Waiting to Depart
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	2524
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	2384
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1334

Tests 

	Magnets under Test from 10-12-2006 to 16-12-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR2363 
	[image: image41.png]



	


	30-11-2006 
	58 h.
	81 h.
	66 h.
	16 h.
	56 h.
	13-12-2006 
	277 h.
	Stripping
	
	

	2
	TBB2
	SSS525 
	[image: image43.png]



	


	05-12-2006 
	97 h.
	-19 h.
	124 h.
	13 h.
	30 h.
	15-12-2006 
	245 h.
	Stripping
	
	

	3
	TBC1
	SSS368 
	[image: image45.png]



	


	05-12-2006 
	29 h.
	17 h.
	40 h.
	11 h.
	76 h.
	12-12-2006 
	173 h.
	Stripping
	
	

	4
	TBC2
	SSS365 
	[image: image47.png]



	


	29-11-2006 
	79 h.
	78 h.
	14 h.
	6 h.
	115 h.
	11-12-2006 
	292 h.
	Stripping
	
	

	5
	TBD1
	SSS674 
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	05-12-2006 
	86 h.
	14 h.
	63 h.
	11 h.
	13 h.
	13-12-2006 
	187 h.
	Stand By
	
	

	6
	TBD2
	SSS694 
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	28-11-2006 
	60 h.
	107 h.
	50 h.
	20 h.
	74 h.
	11-12-2006 
	311 h.
	Stand By
	
	

	7
	TBE1
	MBAR2415 
	[image: image53.png]



	


	07-12-2006 
	39 h.
	19 h.
	15 h.
	14 h.
	20 h.
	13-12-2006 
	107 h.
	Stripping
	
	

	8
	TBE2
	MBBR2417 
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	09-12-2006 
	42 h.
	23 h.
	17 h.
	16 h.
	52 h.
	16-12-2006 
	150 h.
	Stripping
	
	

	9
	TBF1
	MBAR1414 
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	30-11-2006 
	40 h.
	101 h.
	21 h.
	16 h.
	17 h.
	10-12-2006 
	195 h.
	Stripping
	
	

	10
	TBF2
	MBAR2413 
	[image: image59.png]



	


	01-12-2006 
	74 h.
	70 h.
	21 h.
	14 h.
	39 h.
	11-12-2006 
	218 h.
	Stand By
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR2363
	12031.3 A.
(D1_U)
	12849.4 A.
(No Data)
	12850 A.
[image: image61.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS525
	11050.4 A.
(QI_34)
	12226.8 A.
(QE_34)
	12477.7 A.
(QE-34)
	12623.5 A.
(QE_34)
	12750 A.
(QE-34)
	12832.5 A.
(QI-34)
	12850 A.
[image: image62.jpg]



	-
	-
	-
	-
	-
	-

	3
	SSS368
	10989 A.
(QI-12)
	11126.6 A.
(QE-34)
	12223.2 A.
(QE_12)
	12406 A.
(QE-34)
	12525 A.
(QE_34)
	12684.6 A.
(QE_34)
	12850 A.
[image: image63.jpg]



	-
	-
	-
	-
	-
	-

	4
	SSS365
	12541.8 A.
(QE_34)
	12585.3 A.
(QI_34)
	12750.7 A.
(QI-34)
	[image: image64.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS674
	3660.3 A.
(Int_MQM)
	3838.8 A.
(Ext_MQM)
	3927.2 A.
(Int_MQM)
	4128.2 A.
(Int_MQM)
	4216 A.
(Int_MQM)
	4311.2 A.
(Ext_MQM)
	4374.9 A.
(Ext_MQML)
	4406.1 A.
(Int_MQM)
	4434.4 A.
(EXT_MQM)
	4453 A.
(Ext-MQM)
	4471 A.
(I-MQM)
	4485 A.
(E-MQML)
	4504.6 A.
(I_MQML)
	4551.5 A.
(INT-MQM)
	4572.7 A.
(Ext_MQM)
	4604.1 A.
(Int-MQM)
	4626 A.
(E-MQTL)
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	6
	SSS694
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	

	7
	MBAR2415
	11336.5 A.
(D1_L)
	12169.7 A.
(D2_U)
	12765.8 A.
(D2_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	

	8
	MBBR2417
	11422 A.
(D1-U)
	11996 A.
(D2-U)
	12584 A.
(D1-U)
	12850 A.
[image: image68.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	

	9
	MBAR1414
	11066.9 A.
(D1_L)
	11077.6 A.
(D2-U)
	12420.1 A.
(D2_L)
	12690.3 A.
(D1_U)
	[image: image69.png]



	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	

	10
	MBAR2413
	11944.6 A.
(D2_L)
	12840.8 A.
(D2_U)
	12850 A.
[image: image70.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday_13_A
	TBD1
	SSS674 : Resistance values were slightly out of range in PT13.1.
	MTM
	
	Guy was informed. He did the test again and informed Op team not to open NC and upload the file as it is.


	Wednesday_13_N
	TBB1
	SSS519: This MQ is an MQ without a heat exchanger and has to be tested at 4K. No training information is available for this case. 
	MTM
	
	Guy and TC contacted. TC will get information from experts to start the training. TC asked meanwhile to perform the other tests. (HV, IAP, hf..) Pierre gave the information Thursday morning. 

	Wednesday_13_N
	TBD2
	SSS695: Leak in the connection (MRB side, beam flange) found during the pumping. 
	MTM
	
	MTM mechanic team changed the O-R joint

	Wednesday_13_N
	TBF1
	MB1345: While carrying out PT1-Part-III,Ac voltage of 8 volts is being feed. This voltage is not increasing beyond 1 Volt P-P. OP tried to change the pulse generator, change cable, verify of all the connections, but not success.
	MTM
	
	Guy contacted.

	Thursday_14_N
	TBA1
	MB2411: HV test failed at warm. Water in the CFB?
	MTM
	
	

	Friday_15_N

Sunday_17_A
	TBB1
	SSS519: Long training of MQ at 4.5 K (no exchanger). Ultimate not reached after 9 quenches. OP waiting fir TC instruction (ROBTC?)

Magnet reached 11000 A after 20 Quenches.


	MTM
	
	TC and Andzrej advised to continue the training up to I=11000 A.

	Friday_15_N
	TBB1
	SSS519: PT8 not defined  for this MQ at 4.5 K

PT8 failed 2 times: quench in the clamp after several minutes.

Same problem Sunday at the end of the training. MQ did not sustain 11000 A for 10 min, quench n the clamp
	MTM

ACR


	
	TC advised to do PT8 at 10500 A.

Problem of cool down?

TC asked to warm_up the magnet.

	Saturday_16_M
	All
	Cryo OK lost.
	ACR
	
	Flow meter down in one pumping unit. Cryo team worked with the second group.

	Saturday_16_M
	TBB2
	SSS516: Persistent trigger due to V_clamp_E this is continuing since 5AM morning. Hence EF clamp connection could not be done.
	ICS
	
	Guy came and found that

- cable connection for the test was wrong

 +EF-V clamp Vtap was inverted.

	Saturday_16_N
	TBE1
	MB2402: HWREC error message: error in config files.

Problem with IAP : lf application generating 2 sets of files. 
	MTM
	
	OP shift leader contacted Guy. Guy said that he did not understand the origin of problem. He advised to wait 1 hour and re-start the test.

OP shift leader reboot the SUN station and the application was unblocked.



	Sunday_17_M
	TBA1
	MB2411: Shaft in Ap1 cannot be introduced.
	ICS
	
	TC contacted. He advised to perform the other tests and wait for MM Monday.

MTM mechanics found that the TRU was not aligned correctly by ICS.

	Sunday_17_M
	TBD
	SSS669 and SSS695: Inversion the bench name in the display for the on-going tests.
	
	
	OP team will ask Guy to correct the sequence.

	Sunday_17_M
	TBA1
	MB2411: still no cryo OK with this magnet
	ACR
	
	

	Sunday_17_M
	TBB1
	SSS519: Ultimate current reached for MQ at 4.5 K but quench in bus-bar
	ACR
	
	TC asked to repeat the quench (cryo difficulty to cool down the clamp region)

Quench repeated: same results. TC asked to warm-up the magnet.

	Sunday_17_A
	All
	Problem with e-log book. The earlier entries sometimes get deleted.


	
	
	?

	Sunday_17_N
	TBE2
	MB2394:  Capacitance Error during HV at warm Dipole-QH vs  Gnd. Other steps are ok


	MTM
	HV
	Guy called. QH investigation sequence all OK. Guy checked the IFS box and all connections. The Dipole-QH vs Ground still with capacitance problem. Another rack has been tried without success.

Guy waited for Richard consultation Monday.

	Sunday_17_N
	All
	Cryo team informed that due to pumping problem 1.9K temperature is not possible (second unit of pumping was stopped because of suspect noise). So no training possible on 1.9K Magnets.
	
	
	Monday morning: intervention to repair the flow meter of the pumping unit 1.

	Monday_18_A
	TBE2
	MB2394: During HV test MB vs ground displays a high leak current.

	
	
	Richard asked the cryo-team to purge the CFB circuits to remove the humidity. Still the problem.

Richard asked to cool down the magnet.
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